

Review Article

Range-Anxiety Reduction Strategies for Extended-Range Electric Vehicle

Shana Lakshmi Prasad 🕞 and Abhishek Gudipalli 🕒

School of Electrical Engineering, Vellore Institute of Technology, Vellore, India

Correspondence should be addressed to Abhishek Gudipalli; abhishek.g@vit.ac.in

Received 17 July 2022; Revised 23 December 2022; Accepted 24 March 2023; Published 18 April 2023

Academic Editor: Shady Abdel Aleem

Copyright © 2023 Shana Lakshmi Prasad and Abhishek Gudipalli. This is an open access article distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Transportation is the leading source of logistics, and people contribute the significant primary emissions for increasing global warming, so we should avoid such a situation and concentrate on zero emissions. The range of the vehicle is the major problem (range-anxiety) in electric vehicles. Nowadays, researchers focus on range extender optimization since range extenders significantly improve the range of the vehicle with an auxiliary power unit (APU), which can prove consumer satisfaction. However, range extenders can recover energy by proposing the various configurations and systems of extended-range electric vehicles (EREV). Many industries and researchers summarize these efforts to optimize and find the solution for range-extenders. This paper reviewed the most suitable technologies for energy recovery and state-of-the-art topological constructions. The analysis of the study primarily concentrated on optimizing the cost of the systems, fewer emissions, and more fuel economy. The research is based on the range-extenders evaluation and summary for electric vehicles to guide automakers and researchers to invent other state-of-the-art configurations and topologies to obtain optimized EV ranges, namely, enhanced functionality and fewer emissions. A survey on specific extend-range electric vehicles and classification of EREV technologies, the performance of APUs, different control strategies of energy management in EREVs, and various existing difficulties are discussed and further enhanced for the future researcher EREV forecast offered by advanced technologies.

1. Introduction

Road transportation accounts for over 75% of total energy loss in the transportation industry; people and logistics can move around more efficiently with a well-designed transportation infrastructure [1]. The automotive industry has played a significant part in building human civilization, and its economic prosperity has had a long-term impact on pollution levels [2]. Internal combustion engines (ICE) are responsible for 20%–30% of total greenhouse gas emissions (Figure 1(a)) [5]. The greenhouse effect caused by the internal combustion engine uses the ideal gas law, which significantly impacts human health by releasing harmful gases NO₂, CO₂, CO, and NO [1]. Friction and heat loss on the moving component cause ICE vehicles to lose energy. Figure 1(b) shows ICE's energy flow diagram [4]. In this sense, replacing internal combustion engines raises the priority of electric vehicles (EVs) as the best option for emission reduction [6].

In 1834, an earlier development of electric vehicles ran by tricycle power; however, ICE vehicles now account for the majority share in the market [8].

The propulsion of battery-powered electric vehicles necessitates a large battery pack with a limited driving range [9]. Creating a hybrid electric vehicle, which combines an internal combustion engine with an electric motor, is best to increase power and reduce pollution [10]. Hybrid electric vehicle (HEV) promises higher fuel efficiency than ICE vehicles and preserves their state of charge (SOC) during the journey [11]. They achieved zero emissions by utilizing advanced, highly efficient power train technologies, the battery, and hybrid electric vehicles [12]. As a result, the automobile industry chooses to shift to EV power trains; based on this, the shipment progress is rising in the EV and



FIGURE 1: (a) Greenhouse gas emissions sector according to the European environmental agency (EEA) inventory report 2019 [3]. (b) Energy flow diagram of an internal combustion engine [4].

HEVs, with projections based on the International Energy Agency (IEA) report ensuring significant growth as shown in Figure 2 [2, 7].

Regenerative braking and the fuel system are critical issues with HEVs. Plug-in hybrid electric vehicles address this problem, with the battery pack charged via external outlets [13]. Unlike HEVs, plug-in hybrid electric vehicles (PHEVs) allow for an electric motor as the primary mode of propulsion, with an ICE as a backup. When the battery SOC exceeds the limit, the PHEV becomes an HEV with ICE as the primary power source [14]. PHEVs solve the problem of limited driving range while also providing V2G capability [10]. However, due to limitations in energy storage technologies, battery electric vehicles (BEVs) cannot provide consumers with a good driving range [15]. The demand and necessity of range extenders play a significant role in longdriving range electric vehicles with a secondary onboard auxiliary power unit and rechargeable battery pack to overcome range-anxiety issues by providing a continuous supply throughout the trip. The electric motor drives the EREV wheels. APUs typically use a fuel combustion engine and a generator (a generator that includes an integration with the starter is familiar as an ISG). Still, recent investigations have shown better performance in producing electricity with the hydrogen (H₂) fuel cell system [16–19]. A higher degree of electrification and better control strategy is designed and available in EREV due to the compact size of APU compared to the series of plug-in hybrid electric vehicles [20]. Advanced forms of APU, in particular, are being investigated based on energy theory. The gas turbine system is developed by Bou Nader et al. Various comparisons with other type of electric vehicle, Provides in [21-23].

The final output of ISG transfers electrical energy from mechanical torque to chemical energy. An APU will act as a bypass auxiliary device and provide a balanced power supply when the battery SOC is below a certain point. Sometimes, a series of hybrid electric vehicles with the plugin behaves as an EREV by meeting all the power demand requirements [24, 25]. Engine configuration, energy storage system (ESS) charging logic, power train structure, and relative battery size, vary for instance [26–33]. Table 1 shows the powertrain of different vehicles with variants. Lately, in the electric vehicle industry, EREV is one of the essential platforms [37, 38]. The electric vehicle manufacturers have implemented an application for EREV, and products such as the Chevrolet Volt, Nissan e-Power, and BMW i3 REx have gotten much attention [39, 40]. The extended van Ford Transit hybrid range provides a reliable alternative for city driving logistics [41]. According to research, an energy management system depends on the capacity and size of an energy storage system of an EREV [42]. The battery capacity depends based upon the selection of the appropriate engine. The EREV's operating characteristics, power handling capacity, energy management, and performance depend upon every component, to maintain balanced power by adjusting the objectives of dynamic programming [43–45].

Recent research has focused on APU charging control optimization and EREV energy management [46–50]. For instance, some researchers discovered that rule-based

methods are unsuitable for various driving patterns used in the control system; alternatively, optimal solutions are considered for Energy Management System (EMS) [31, 51–53]. These issues can improve or be solved with a few references. This paper shows power management and APU charging control strategies for electric vehicles.

The remaining paper organizes the following sections. Section 2 introduces the electric vehicle technology with the state-of-the-art extended-range electric vehicle. However, Section 3 provides the technological comparison of EREV. Furthermore, Section 4 discusses the widely used control strategies and methods compared to EREV. Furthermore, Section 5 then delves into the APU charging control strategies for EREV energy management in depth. Section 6 examines a case study of a typical EREV model. Finally, this paper concludes with Section 6. Modern EREV structures and novel optimized EMSs are the future development of knowledge and inspiration.

2. Electric Vehicle Technology with the State-ofthe-Art Extended-Range Electric Vehicle

The range extender (RE) consists of an auxiliary power unit (APU) and can be operated with a satisfactory solution to extend the EV's autonomy. An engine with an external or internal combustion engine and a generator are the primary components of the range extender (RE). The main function of a RE is to improve the mileage of electric vehicles. If the battery's state of charge (SOC) is below certain boundary conditions in the operation of the driving mode, the vehicle operation is continuous throughout by recharging the vehicle's battery. The significant comparison between the PHEV and EREV is that the motor always propels the wheels. In the depletion mode of operation or to drive the vehicle, the generator recharges the battery [54]. An APU is configured with the series configuration and will act as the primary system. The interconnected subsystems are the electric motor, electronic management, battery, and generator. To maintain the optimal functioning of the controller in electronic and mechanical power obtained from the battery and the electrical energy converted from the electric motor, the EREV consists of two basic modes of operation, namely, an extended-range mode of operation and a pure electric vehicle, where the distance is longer for an extended-range electric vehicle, and the distance is shorter for a pure electric vehicle mode of operation.

In the vehicle, the SOC of the battery is below a certain level, and the pure electric vehicle mode of operation is continuous. The function of RE depends upon certain boundary conditions of the battery and the functioning maintained by the power manager.

An energy flow configuration as shown in Figure 3(a) charge sustain period and (b) depleting period, forthcoming EVs overcome various social and technical challenges over ICE vehicles. Compared to the ICE vehicles, the electric vehicle's driver faces range-anxiety problems due to the shorter driving range. The range-anxiety stock is limited in the present batteries (0.565 MJ/kg–Li-ion



FIGURE 2: Global EV/HEV shipment estimates [7].

TABLE 1: Comparison of alternative vehicle configurations with various powertrain models.

Model	ICEV [34]	Parallel [20]	Series-parallel [28]	EREV [34]	BEV [35]	Series [20]	PHEV [36]
Range	$\sqrt{}$	$\sqrt{}$	$\sqrt{}$	$\sqrt{}$	\checkmark	$\sqrt{}$	$\sqrt{}$
Performance	$\sqrt{}$	\checkmark	\checkmark	$\sqrt{}$	$\sqrt{}$	\checkmark	\checkmark
Electric-motor	0	1	2	2	1	2	2
Battery capacity	—	\checkmark	\checkmark	$\sqrt{}$	$\sqrt{\sqrt{\sqrt{1}}}$	$\sqrt{}$	$\sqrt{}$
Engine efficiency	\checkmark	$\sqrt{}$	$\sqrt{}$	$\sqrt{\sqrt{\sqrt{1}}}$	—	$\sqrt{}$	$\sqrt{}$
Power train complexity	\checkmark	$\sqrt{}$	$\sqrt{\sqrt{\sqrt{1}}}$	\checkmark	\checkmark	\checkmark	$\sqrt{}$
Transmission of engine	1	1	1	0	0	0	1

battery); fossil fuel energy density is compared to (43.48 MJ/kg) [55].

2.1. Technological Categorization of EREV. An electric propulsion system plays a vital role in an EREV. The main parts are the propulsion (or) transmission system, wheels, and the electric motor drive. The classification of electric motors is of three types, namely, in-wheel, alternating, and direct current. The significant following are the needs of an EREV motor:

- (1) The density of power and power instant is very high.
- (2) The consumption power is very high at high speeds and then cruising. For low speed maintenance, starting and climbing need high torque.
- (3) Constant torque and constant power regions achieve a substantial range of speed.
- (4) Torque response is very high.

- (5) A range of torque and significant speed obtain high efficiency.
- (6) The dynamic operating conditions of the vehicle achieve high robustness and more reliability.
- (7) The cost of the motor is reasonable.

2.1.1. Extended Range Internal Combustion Engine (ER-ICE). The main parts of the range extender are a permanent magnet synchronous generator, an internal combustion engine, and a fuel tank [17, 46, 55–82], as shown in Figure 4. This type of structure is unsuitable for coupling between the RE and wheels. Only the vehicle output power fills the need to drive and needs a solid cut-off point to meet the traction characteristics of the vehicle. On the other hand, RE is to maintain the region of high frequency [80]. Another alternate solution is that many studies concentrated on reducing pollution levels, such as diesel [83] and natural gas [25].



FIGURE 3: An EREV energy flow configuration; (a) charge sustaining period; (b) charge depleting period.



FIGURE 4: A simple block diagram of ER-ICE configuration.

2.1.2. Extend Range Regenerate Shock Absorber (ER-RSA). The suspension of the vehicle's primary component is a shock absorber to filter the vibrations by using suspension springs on rough roads. The heat and hydraulic friction absorb the energy from vibrational sources by shock absorbers [84]. At present, a wide variety of three RSAs are available. The linear or rotary is the scheme in operation [85]. In the first method, electric power is generated by an electromagnetic used directly. An electromagnetic induction

principle will work on these to convert the kinetic energy of vertical oscillation to electricity by an RSA of linear electromagnetic.

The RSA of hydraulic is the second method. This RSA can be powered by using an oscillatory motion to run an electric power generator. Modern shock absorbers utilize oil to flow in a parallel oil circuit by replacing the existing hydraulic shock absorbers. These fluids drive the DC/AC generator parallel to the hydraulic motor [86].



FIGURE 5: A simple block diagram of ER-RSA configuration.

The RSA of mechanical is the third method. It is the most efficient and has a high average power [84]. The combined architecture of RSA with the supercapacitor increases the battery lifetime enormously. The major parts are as follows: (1) the suspension vibration as input block, (2) the transmission block, (3) the generator block, and (4) a block with power storage, as shown in Figure 5.

2.1.3. Extended Range Regenerative Braking (ER-RB). The electric vehicle deceleration process leads to transmitting torque to the axle and simultaneously incorporates the charging of the battery, as shown in Figure 6. Under this condition, a kinetic energy system represented by a regenerative braking system (RBS) enhances the overall energy efficiency of an electric vehicle (EV). Two modes of the braking system are used in EVs to produce the brake torque together or in either form, namely, regenerative braking and friction braking of the system [87]. A suitable control strategy is needed to improve energy recovery and safety and reduce the loss rate. A revised regenerative braking control strategy (RRBCS) reduces the inefficiency and prolongs the battery life by increasing the maximum energy recovery and safety [88].

A transmission system with high transmission efficiency needs to improve. A hybrid braking method can improve transmission efficiency using an automatic mechanical transmission (AMT). By applying the braking force to the transmission gear system, in this condition, the speed does not affect. However, the vehicle's condition changes by the braking force and transmission gear accordingly; this method enhances the EV system's economy. Further improvement of this method needs to operate in the maximum permissible condition for maximum energy recovery. A suitable control strategy is required to improve the rate of energy recovery, stability, and safety [89].

2.1.4. Extended Range-Fuel Cell (ER-FC). A fuel cell is a device that converts directly into electrical energy and heat from the electrochemical energy conversion [90]. It incorporates to produce water and electricity by redundant oxidants in the output of the electrochemical process [91].

A fuel cell consists of an anode, cathode, and electrolyte, which are the primary components. In this process, the anode delivers the hydrogen, and oxygen provides the cathode [92]. With an external circuit, the positively charged particles move towards the cathode side, and the electrons or negative-ion charged particles move towards the anode to rejoin. In this process, the moment of electrons creates electricity in external circuits. Then, the combination of positively charged particles and the oxygen collision with the electrons produces heat and water, as shown in Figure 7 [18, 93–99].

2.1.5. Extended Range Microgas Turbine (ER-MGT). Microgas turbines (MGT) for small output power are in the range of 30–500 KW [100]. The major components of the MGT are a radial compressor with a single stage, a recuperator, and a radial turbine section. Four types of the process consist of an MGT cycle as follows:

- (1) The radial compressor consists of inlet air, and it compresses inlet air
- (2) The hot air mixed with the fuel in the combustion chamber is passed from the recuperator
- (3) In the stages of turbine stages, it expands hot gas and
- (4) In this process, the air compressor drives the equipment (usually the generator), and succeeding mechanical energy develops from gas energy.

The gas turbine, according to manufacturers, represents the most effective future technology. Significantly, alternatives are needed. RE is an alternative to the EV, and MGT is an alternative to the ICE. The gaseous exhaust emissions are minimal, produced by the MGT such as carbon monoxide and hydrocarbons. In contrast, the application is more stable than the ICE. The carbon monoxide level of potential reduces [101–103]. Figure 8 shows that the battery and generator connected to MGT-ER are in a series configuration.

2.1.6. Solar Energy Storage Extended Range (SES-ER). The heat energy and electricity produced by the sunlight then cause the moment to start, flowing electrons to produce electricity in photovoltaic cells (PV cells)—automakers are

International Transactions on Electrical Energy Systems



(b)

FIGURE 6: The energy flow in ER-RB configuration: (a) generator mode; (b) traction mode.



FIGURE 7: An ER-FC configuration block diagram.



FIGURE 8: An ER-MGT configuration block diagram.

interested in solar energy storage (SES) systems for economic, safety view, and cleanliness. Studying systematic and lighter SES has enhanced the analysis of numerous automobile industries [104]. Integration of renewable energy sources is needed to improve the charging capacity of the EREV [105] and also reduce the overall cost-effectiveness by utilizing solar-type renewable energy sources [106] and further required to improve the storage capacity in the electric vehicle [107].

However, the charging infrastructure places a vital role in EV energy management [108]. Furthermore, it enhances the battery charging topologies and widely used methodologies in EV battery technologies [109]. In city driving, the 10% autonomy uses PVC in cars [110]. Ezzat and Dincer [111] offered modern complete energy storage solution for EREV. The proposed method incorporates batteries, PVCs, and a fuel cell (FC). The findings suggested that using solar cells makes EREV's energy storage technology more efficient. The ideal complements to a range extender. Figure 9 shows the layout of an ER-FES.

2.2. EREV Comparisons with Various Technologies. The design consideration and parameter estimation of a rangeextended electric vehicle are the most critical issue in developing the technologies and their comparisons, analyzing different technologies, and analyzing the relationship of their components and configurations described in Table 2. The range extender selection criterion for the vehicle design depends upon the specific characteristics of the vehicle, and these are compared and evaluated by the following:

- (i) Required power from the system;
- (ii) The amount of additional range;
- (iii) The system efficiency is global;
- (iv) Emissions of the system.

3. Extended-Range Electric Vehicle System Design and Powertrain Classification

There are two next-generation vehicles, namely, hybrid electric vehicles (HEVs) and (BEVs). As seen in Section 1, in the battery, electric vehicle storage capacity is the primary constraint infrastructure availability for charging, design rationality of EMS, and charging speed of the battery. The first phase is many social and technical challenges.

Power conversion efficiency is higher in this structure, which requires a high energy storage system (ESS) compared to the other HEVs described in Table 1. An extended-range electric vehicle has proposed a new HEV definition in recent years. Based on long-distance driving ability, it progressed through rapid development and met the government consumption criteria for indices [50]. The best alternatives for future road transportation are widely acknowledged [25, 37, 152]. As shown in Figure 6, an EREV is a simplified topological construction.

3.1. Basic Pattern and Operation of EREV. An EREV consists of a hybrid energy storage system (HESS) with two operating modes, namely, charge-depleting and sustaining methods. The charge depleting mode maintains till the dead zone of



FIGURE 9: An ER-SES configuration block diagram.

SOC. In this mode, the ICE is disabled, and the sustaining charge mode supports the remaining SOC. This way, charge-depleting (CD) mode is reliable with less power consumption and more SOC maintained by the operating modes of a hybrid energy storage system. In contrast, if the state of charge (SOC) falls below a certain level or the power demand increases, the engine's lack of power is compensated. In this case, the APU's electric energy can power the ESS and drive the motor via power converters. Furthermore, studies show HESS power management, APU charging control, and power flow distribution [153–156]. The following sections deal with the APU control strategies and power management of HESS.

3.2. The Auxiliary Power Unit (APU). The vehicle design objective dictates the choice of an APU for EREV [157, 158] shown in Figure 10. A conventional vehicle's engine selection is quite different from EREV. Typically, a broad range of dynamic performance requires sufficient power to maintain the required speed range, high fuel consumption, and more significant displacement [37]. On the other hand, the engine selection for APU is different storage. Increasing the degree of electrification reduces the number of pistons and displacement [49]. Specially designed range extenders powered by small engines boost the electric range in EREVs since they have fewer cylinders [159, 160], to obtain minimum designing and manufacturing costs by using small displacement engines consisting of fewer materials.

As shown in Figure 11, electric energy storage will consist of two basic configurations. First, the battery energy storage system is only a part of most energy storage systems. The hybrid energy storage system achieves high peak discharge and extended battery life, and the converter topology maintains the voltage stability between the battery and the supercapacitor [51]. The smaller displacement engines will satisfy traditional engines' manufacturing and design characteristics [37]. Furthermore, the engine does not need to run over a wide speed range. A high-efficiency zone of an engine is possible for the EREV with lesser noise and lower vibrations. This way, the standard transmission is not required for the EREV power train [161, 162]. Compared to the Otto cycle, APU control strategies are higher inherent characteristics. In [159], a specially designed two-stroke engine does electric power generation in a single cylinder. In the new EREV project announced by Mazda, the range extender consists of a redesigned engine [163]. The rotor reliability and stability are maintained by the boxer engine perfectly.

3.3. Hybrid Energy Storage System with Various Energy Storage Elements. Figure 11 depicts their samples. The most basic ESS configuration is a battery-powered storage system. HESS compensates for high peak discharge, extends the battery, and reduces voltage stability by combining SC and battery using a buck-boost converter [51]. The following are the characteristics of the battery and SC.

3.3.1. Battery. EREV, like other EVs, stores electric energy as a rechargeable battery. Lithium-ion (Li-Ion) batteries and nickel-metal hydride (Ni-MH) are two types of batteries used in modern electric vehicles. Ni-MH batteries can be recharged up to 2500 times when charged to 80% depth of charge [164].

The greater energy density and high efficiency have the Li-ion batteries [165]. A strict charging control will apply constant current charging, usually regular voltage, to charge the Li-ion batteries. To simplify, we calculate the SOC of a battery pack [24], where U_{oc} is the open-circuit voltage, R_r is the internal resistance, and P_{batt} is the battery terminal power. Q_{bat} is the nominal battery capacity [24]. The Li-ion battery, charging, and discharging protect the battery from the estimate of SOC.

$$SOC = \frac{U_{OC} - \sqrt{U_{OC}^2 - 4R_r P_{batt}}}{2Q_{batt} R_{tml}}.$$
 (1)

3.3.2. Supercapacitor (SC). The cost is the main primary effect; installing the batteries to reduce this effect is replaced by the inherent properties of supercapacitor (SC). Compared to Li-ion batteries, the supercapacitor had a longer life span and higher power density [155, 166] in the 1990s. The Mazda model's first supercapacitor application was in automotive powertrain applications to boost the restoration of retrieved energy from regenerative braking power [167]. Due to the

		b		
Extended range of the system	Extra range	System power	Emissions	Efficiency
ER-RSA	Can power an 8W lidar for 323 days or a 2W camera for 1292 days [112]	8–40W [112] 0.74–0.78 kW [113] 19.2–67.5 W [115] 4.3 W [84]	No	70-80% [112] 71-84% [114] 33-63% [115] 87% [84] 16% [84]
ER-TAE	80% fuel consumption savings [116]	710 W [117] 1029 W [118] 58 W [114] 1.5 kW [114]	Low	33.8–38.7% [116] 30% [117] 5.4% [118] 18% [119] 16% [114]
ER-FES	50% mileage over [112] 1.17% mileage over [121]	40 kW to 1.6 MW [120] 60-101 kW [112] 1-20 kW [123]	No	60% [112] 90–95% [122] 70–90% [124]
ER-WT	Add up to 10% [125] 7.27 km [126]	2.64 kW [126] 0.1–1.1 kW [127]	Low	75% [126] 75–90% [127]
ER-PVC	19.6 km [128]	68.2–300 W [128]	No	91.2% [128] 20.2-23% [128]
ER-MGT	370 km [129]	32 kW [130] 100 kW [132] 63.3 kW [133]	Low	47.2% [131] 28% [130] 30% [132] 35% [129] 38% [133]
ER-RB	32.1–47.7% of the total recoverable energy [134] 1.18% SOC improvement in SOC [136]	14.8kW [135] 55.75-82.66kJ [134] 298.75kJ [136]	No	79–94% [135] 30–60% [76] 47% [134]
ER-rotary engine	80 km [137] 321 km [140]	3.8 kW [138] 20 kW [137]	Low	73% [139] 78% [138] 77% [137]
ER-fuel cell	500 km [141] 650 km [142] 665 km [144] 594 km [144] 1500 km [146]	20 kW [141] 85–83 kW [119] 1200 W [93] 25 kW [145] 128 kW [142]	No	70% [141] 63.6-72.4% [143] 43% [93] 55.21% [145]
ER-ICE	232.79% [79] 430 km [77] 51–139 km [25] 380 km [149] 330 km [151] 676 km [151]	30 kW [77] 35 kW [78] 5.5 kW [25] 111 kW [150]	Low	20–40% [147, 148] 31% [77]

TABLE 2: An extended range of system compared with various range extender technologies.

10

International Transactions on Electrical Energy Systems



--- Electrical Connection

FIGURE 10: Overview of engine based EREV power flow.



FIGURE 11: General diagram of an ESS and HESS; (a) ESS; (b) HESS.

lack of chemical reactions, a supercapacitor has a greater power density than a battery by charging its electrodes [168]. Supercapacitor is most suitable for fast charging and discharging applications due to its inherent characteristics, high efficiency, and low internal resistance than the battery [169]. Regenerative braking occasionally delivers a high current to the ESS, causing the battery to be damaged [170]; installing an SC can prevent this damage [169]. An SC's energy calculation equation is as follows:

$$E = \frac{1}{2}C(V_2^2 - V_1^2), \qquad (2)$$

where *E* denotes total useable electric energy, *C* denotes electric charge, V_2 denotes charged terminal voltage, and V_1 denotes cut-off voltage. We can see that an SC's total electric energy is proportional to its terminal voltage.

Various methods are presented in this literature to compute the supercapacitor's state of charge (SOC) [171]. Considered to be the SOC of SC is a linear relationship between calculated SOC by equation (3) with open-circuit voltage V_{SC} as follows:

$$SOC_{SC} = \frac{V_{SC}}{V_2}.$$
 (3)

Because 75 per cent of the useable SC energy *E* is released when the SOC falls to 0.5 [171], SOC should be able to work within a specific range [0.5–1]. Alternatively [172], the current integral approach equation gives the SOC of SC in real-time:

$$SOC_{SC}(t_0) = SOC_{SC}(t_1) + \frac{1}{E} \int_{t_0}^{t_1} I_{SC}(t) dt.$$
 (4)

4. Energy Management Control Strategies of EREV

Research topics for the EREV energy management system include control strategies for auxiliary power units and designing HESS for dynamic power management. Using SC, the auxiliary power unit can operate in high-performance mode; depending on the APU control pattern, the HESS size will also differ [173].



FIGURE 12: General diagram of a power flow strategy.

4.1. Charging Control Strategy of APU. Based on simulation and experimental findings [36, 52, 174–176], the maximum efficiency of the Proton Exchange Membrane (PEM) fuel cell and ICE is about 66% and 41%, correspondingly. The output torque and speed determine the efficiency of APU with engine-based one [177]. In contrast, a fuel cell unit's efficiency is determined primarily by its size, fuel purity, and working temperature [178]. Auxiliary power unit charging control strategies have power flow and optimal range strategies.

4.1.1. Strategy for Power Flow. Figure 12 shows the simplified diagram for the power flow strategy, even with the flexible fuel cell discharge behavior used on extended-range electric vehicles with fuel cells (FC) [179–182]. In the case of APUs powered by a combustion engine, a comprehensive, high-efficiency operating range is necessary for the engine [46, 183]. Furthermore, the intermittent power imbalance between the actual power consumption and the APU output must be considered [46, 184]. The combination of SC and HESS has minimized the errors and enhanced the power handling capacity [185].

Furthermore, a quick control convergent speed for this strategy, the driving requirements of APU, and the fuzzy logic control fulfil the power demands [186]. As a result, the fuel cell bus is a 12-hour real-world driving cycle used to estimate power requirements for HESS. The results demonstrated that a small fuel cell APU may offer the majority of the power to drive as the battery's state of charge remains high [182]. The fuel cells prior to power selection needed then thereby required charging approach of APU. On the other hand, the fuel cell efficiency achieved more than that 43% by selecting a fuel cell power range (about 65–180 kW) with maximum reliability [187]. However, the combination of ISG and combustion engine used, as APU, will give rise to high fuel usage.

This author created a slot-PM-aided generator under variable conditions with a combination of a combustion engine and a generator. The simulation outcomes revealed that the DC-excited structures are being lesser power density when compared to the slot-PM-aided generator, almost double [188, 189]. An energy management design is based on an adapted online drive cycle/power predictions by introducing the approach of Through-The-Road (TTR). High efficiency depends upon the total power consumption of APU, followed by torque [100, 122]. While protecting the end-state SOC, which could minimize the energy input consumption, the hardware-In-Loop (HIL) test disclosed that hydrogen consumption in the traditional powerfollowing method is 8.99 per cent more than in the online strategy.

Furthermore, these authors used various strategies; in the charge sustaining (CS) stage, the battery and APU power distribution have been selected. Different methods were applied using pseudospectral optimal control (SOC) and DP [46]. The efficiency of energy improvements is carried out in [17, 50]. An ideal economic curve was selected by an engine, in the selection process, which provides the power to the powertrain by an APU.

4.1.2. Strategy for APU Optimal Range. A strategy in good range maintains APU with high fuel efficiency. Energy conversion efficiency is more if every engine supports the optimal range. As a result, this strategy configures the engine to operate at all times within an operating range or some fixed optimized points. For example, we consider the auxiliary power unit in [24]. High efficiency maintains by three operational points; the backup power is used to charge the ESS by turning on the APU. Figure 12 [153–156] depicts the appropriate engine operation range. Before selecting, a logic diagram shows how to choose the proper engine operation range using this strategy in Figure 13 [189]. To find out the battery condition and the transient power required, we perform calculations of a series type. The best range is possible by limited engine operation and a nonlinear charging strategy.

Scholars now use machine learning and mathematical approaches to develop range optimization strategies [190]. The system can adjust to an uncertain operating



FIGURE 13: Logic diagram for optimal control strategy [189].

environment using machine-learning methods. A genetic algorithm was used in [191]. A genetic algorithm will increase engine efficiency by training and testing data based on injecting time optimization on a diesel engine.

To investigate the efficiency of a 3-cylinder Atkinson engine using an artificial neural network for APU charging, optimized engine operating points led to a high efficiency of 40.2 per cent. According to the results [135], in [192], the distance of the gas station and battery SOC are two inputs trained and led by a neural network controller design [193] to predict the engine-battery interactions and drive behavior by stochastic model predictive control (MPC). The driving style and traffic conditions influence the engine-battery power ratio. Furthermore, the optimal range strategy depends upon the PSOC and DP algorithms [46]. Here, the object is used as the actual power demand.

4.1.3. Comparison Analysis of Two Strategies. The powerfollowing approach minimizes the power error between vehicle power required and APU output by reducing the size of an EREV and ESS. Because of the current research activities on these strategies in the APU charging control area and developing real-time experiments with HIL, there may be an opportunity to create new methods. In combination with HESS shown in Figure 14 [153], the engine or fuel cell of the vehicle will start to output power as it reaches its best performance levels. The required drive power for an EREV discussed is as follows:

The strategy must be redesigned [154]. Furthermore, the hybridization of HESS and APUs must achieve this level of performance. It aims to combine many more modern HESS architectures for power management in [194]. A HESS power management designed with three system-level objectives in this article as follows:

(a) To improve the battery life span by reducing the excessive amount of charging/discharging; (b) by utilizing the fast discharging and charging capability of the supercapacitor; (c) HESS sizing should improve to reduce weight and cost. As a result, a multiobjective optimization problem can be solved more effectively with dynamic power management classifications. Using advanced machine learning methods for continuous improvement will be necessary for the future of functional HESS power management. The operation requires multiple input signals. For example, reinforcement algorithms such as PPO, DPG, and DDPG have received widespread attention [195, 196], and HESS can provide continuous signals for data acquisition during training.



FIGURE 14: Widely used HESS topology structure [153].



FIGURE 15: The concept of the Chevrolet volt-based model [197].

5. A Specified EREV Case Analysis

The Chevrolet Volt chosen has been on the market for over a decade in this case, because it is one of the most representative EREVs. Initially, the prototype is Figure 15. The first-generation Chevrolet Volt had a 55 kW ISG, a 63 kW engine with an APU, and a 16-kWh battery pack, resulting in a maximum distance of travel of 620 km. As an outcome, the peak trip length is 620 km [198–200]. The maximum trip range was achieved to 680 km by the added features of the second generation with the compact power coupling system, a new coupling planetary gear, and a slightly large battery pack [200]. In an electric mode of operation, the traction motor delivers a power output of 111 kW [194].

5.1. On-Road Testing and Validation. Governments and agencies have proposed some initiatives to assess the capability and relevance of long-range electric vehicles, and the Chevrolet Volt was tested on the road. Energy consumption

and travel range are the test subjects. A Chevrolet Volt can travel 55 km on pure electric power alone (163 miles), which is significant. This strategy enables the APU to be useful for longer travel distances through energy management. A comprehensive investigation of charging and driving for 923 test targets across 7.6 million kilometers was conducted at Idaho National Laboratory in 2013 [199]. Among the studied sample, the average daily travel distance was 65.5 km, and the average number of charges was 1.46. As a result, the Chevrolet Volt can only cover a travel distance of 73% in the EV mode if the vehicle leaves the garage with a fully charged battery.

APU will turn on the extended-range mode when the Chevrolet Volt's battery SoC falls under 17% [185]. By default, the Chevrolet Volt's charging strategy is to charge the APU and battery by utilizing the remaining energy to charge the APU. To a higher level, the battery will receive all its power [201]. The new planetary system in the Chevrolet Volt's second generation enables a new extended-range



FIGURE 16: Shows Chevrolet volt Wh/km and average travel distance for different seasons [204].

TABLE 3: Various energy storage performance and a specification comparison table with the effect of cost.

Energy element	Efficiency of energy (%)	Lifespan (cycles)	Energy specification (Wh/kg)	Market price (US\$/kW)	Density of power (W/kg)	Literature
PEMFC	45-66%	2500-5000 hrs	_	40-53	1000-1600	[172, 177, 178], [208-210]
Li-ion	85-95	2000	180-300	100-150	1000-3000	[211-213]
SC	>95%	1M	4.1-6.0	33.5-43.6	10,000-14,000	[166, 169, 205, 206]

mode. To improve power train efficiency by linking with the engine and ISG, validation of the model in [185] revealed that the cut-off fuel of the engine was only when the vehicle speed exceeded 65 km. By restarting the engine at higher speeds, consumption of energy is very low. According to the experiment results, the engine only operated at 25 to 35 kW.

For an EREV to be energy efficient, its operating temperature significantly impacts [197, 202]. Several studies have been conducted on the Chevrolet Volt at low temperatures to study its impact [202]. Engineers tested the Chevrolet Volt in Idaho to see how the cold climate affected the APU and discovered that the APU operated regardless of the ambient temperature when the battery SOC was below -2.8° C.

They also discovered that the cabin temperature varied based on the climate control system, caused by the average temperature of -26.1° C and the fuel consumption falling below 47 mpg (5 L/100 km). Based on a Canadian environmental study [203], the energy consumed to heat the cabin in winter is much higher than the amount consumed to cool it down in summer. In one driving cycle, CD on-road results indicated that heating the cabin consumed 4.3 kW at 1.6°C, while only 1.4 kW at 36°C. Furthermore, when the APU is switched off, the cabin is conditioned and runs in pure electric range. Figure 16 shows the energy range data on season [203]. The maximum distance is only possible by 103 km at 27°C, which is a slightly higher temperature when compared to the minimum distance of 5°C, which is 41 km, as shown by the results.

The supercapacitor characteristic differs with the same as that of a battery. The power density will be extremely high, ranging from 10 to 14 kW/kg [205, 206]. Because of its high cost and small energy storage capacity, it is not acceptable as a primary source. Nevertheless, the cost analysis constructed between market raw materials and the hybrid energy storage systems [203, 207–209] is changing slowly. In 2018, the lowest price per kilowatt was \$33.50/kW, as shown in Table 3, far less than its early-twentieth-century pricing. As a result, it is still expensive in contrast to the price of Li-ion batteries, the total cost of operation (TCO). It has become more economical to pair a smaller-size SC with a battery pack to conserve the battery state of health (SOH) [166, 169, 205, 206].

6. Conclusion

This paper aims to summarize and address energy management strategies of the electric vehicle, the state-of-art of electric vehicles, and the ranging issues engineers, and researchers have tested and proven the design rationality. First, this paper provided a brief overview of the types of regenerative braking, EREV structure, and energy management, emphasizing various control strategies. APU charging strategies are described here as optimal range and power following, and discussed pros and cons shared these two in detail. The factors influence the APU's charging patterns, vehicle speed, battery SOC, and power command. In addition, it illustrates how manufacturers will easy to implement rule-based strategies. Nevertheless, all other methods that optimize and learn data patterns that use machine learning are particularly inapplicable to commercial vehicles. Essentially, an efficacious power management system is well-designed. In addition to extending the HESS lifetime, it will save installation space and provide adequate power output. The Chevrolet Volt examined its road and energy management and performance. Finally, future EMS development for EREV will improve power train efficiency and overall system stability; novel algorithms and hardware implementation are used to overcome range anxiety and deal with electric vehicle energy management.

Abbreviations

APU:	Auxiliary power unit
BEV:	Battery electric vehicle
CD:	Charge-depleting
CS:	Charge-sustaining
CO:	Carbon monoxide
CO_2 :	Carbon dioxide
EMS:	Energy management strategy
EREV:	Extended range electric vehicle
ESS:	Energy storage system
FC:	Fuel cell
H ₂ :	Hydrogen
HESS:	Hybrid energy storage system
HEV:	Hybrid electric vehicle
HIL:	Hardware in the loop
ISG:	Integrated starter generator
Li-Ion:	Lithium Ion
MPC:	Model predictive control
Ni-MH:	Nickel-metal hydride
PEM:	Proton exchange membrane
PHEV:	Plug-in hybrid electric vehicle
SOC:	Pseudospectral optimal control
SC:	Super-capacitor
SoC:	State-of-charge
SOH:	State of health
TCO:	Total cost of ownership
TTR:	Through-the-road.

Data Availability

No data were used to support the findings of this study.

Conflicts of Interest

The authors declare that they have no conflicts of interest.

Authors' Contributions

Shana Lakshmi Prasad helped in conceptualization, methodology, writing-original draft preparation, visualization, and investigation. Abhishek Gudipalli helped in sSupervision, validation, and writing-Reviewing and Editing.

Acknowledgments

The authors acknowledge the support of the Vellore Institute of Technology, Vellore, India, for this work. In addition, the authors would like to thank the respective copyright holders for permission granted to include cited graphics, images, tables, and figures in this work.

References

- S. F. Tie and C. W. Tan, "A review of energy sources and energy management system in electric vehicles," *Renewable* and Sustainable Energy Reviews, vol. 20, pp. 82–102, 2013.
- [2] S. Chakraborty, H. N. Vu, M. M. Hasan, D. D. Tran, M. E. Baghdadi, and O. Hegazy, "DC-DC converter topologies for electric vehicles, plug-in hybrid electric vehicles and fast charging stations: state of the art and future trends," *Energies*, vol. 12, no. 8, p. 1569, 2019.
- [3] U. Secretariat, Annual European Union Greenhouse Gas Inventory 1990–2009 and Inventory Report 2011 2011, European Environment Agency, Copenhagen, Denmark, 2013.
- [4] K. T. Chau and C. C. Chan, "Emerging energy-efficient technologies for hybrid electric vehicles," *Proceedings of the IEEE*, vol. 95, no. 4, pp. 821–835, 2007.
- [5] S. Reddi Khasim and C. Dhanamjayulu, "Selection parameters and synthesis of multi-input converters for electric vehicles: an overview," *Renewable and Sustainable Energy Reviews*, vol. 141, Article ID 110804, 2021.
- [6] C. Thiel, A. Perujo, and A. Mercier, "Cost and CO2 aspects of future vehicle options in Europe under new energy policy scenarios," *Energy Policy*, vol. 38, no. 11, pp. 7142–7151, 2010.
- [7] M. Hawes, Emerging Solutions to Hybrid and Electric Vehicle DC: DC Converter Design and Test, Keysight Technol, St Rosa, CA, USA, 2017.
- [8] C. Shen, P. Shan, and T. Gao, "A comprehensive overview of hybrid electric vehicles," *International Journal of Vehicular Technology*, vol. 2011, Article ID 571683, 7 pages, 2011.
- [9] G. Fontaras, T. Grigoratos, D. Savvidis et al., "An experimental evaluation of the methodology proposed for the monitoring and certification of CO2 emissions from heavyduty vehicles in Europe," *Energy*, vol. 102, pp. 354–364, 2016.
- [10] K. V. Singh, H. O. Bansal, and D. Singh, "A comprehensive review on hybrid electric vehicles: architectures and components," *Journal of Modern Transportation*, vol. 27, no. 2, pp. 77–107, 2019.
- [11] L. Kumar and S. Jain, "Electric propulsion system for electric vehicular technology: a review," *Renewable and Sustainable Energy Reviews*, vol. 29, pp. 924–940, 2014.
- [12] P. K.-H. Dost, P. Spichartz, and C. Sourkounis, "Charging behavior of users utilizing battery electric vehicles and extended range electric vehicles within the scope of a field test," *IEEE Transactions on Industry Applications*, vol. 54, no. 1, pp. 580–590, 2018.
- [13] A. Mahmoudzadeh Andwari, A. Pesiridis, S. Rajoo, R. Martinez-Botas, and V. Esfahanian, "A review of Battery Electric Vehicle technology and readiness levels," *Renewable* and Sustainable Energy Reviews, vol. 78, pp. 414–430, 2017.
- [14] E. Hossain, R. Perez, S. Padmanaban, L. Mihet-Popa, F. Blaabjerg, and V. K. Ramachandaramurthy, "Sliding mode

controller and lyapunov redesign controller to improve microgrid stability: a comparative analysis with cpl power variation," *Energies*, vol. 10, no. 12, p. 1959, 2017.

- [15] A. Solouk, M. Shakiba-Herfeh, J. Arora, and M. Shahbakhti, "Fuel consumption assessment of an electrified powertrain with a multi-mode high-efficiency engine in various levels of hybridization," *Energy Conversion and Management*, vol. 155, pp. 100–115, 2018.
- [16] R. Álvarez Fernández, S. Corbera Caraballo, F. Beltrán Cilleruelo, and J. A. Lozano, "Fuel optimization strategy for hydrogen fuel cell range extender vehicles applying genetic algorithms," *Renewable and Sustainable Energy Reviews*, vol. 81, pp. 655–668, 2018.
- [17] J. Du, J. Chen, Z. Song, M. Gao, and M. Ouyang, "Design method of a power management strategy for variable battery capacities range-extended electric vehicles to improve energy efficiency and cost-effectiveness," *Energy*, vol. 121, pp. 32–42, 2017.
- [18] X. Wu, Y. Xu, D. Zhao et al., "Extended-range electric vehicle-oriented thermoelectric surge control of a solid oxide fuel cell system," *Applied Energy*, vol. 263, Article ID 114628, 2020.
- [19] X. Wu, X. Hu, X. Yin, Y. Peng, and V. Pickert, "Convex programming improved online power management in a range extended fuel cell electric truck," *Journal of Power Sources*, vol. 476, Article ID 228642, 2020.
- [20] R. M. Dell, P. T. Moseley, and D. A. J. Rand, *Towards Sustainable Road Transport*, Academic Press, Cambridge, MA, USA, 2014.
- [21] Z. Chen, C. C. Mi, B. Xia, and C. You, "Energy management of power-split plug-in hybrid electric vehicles based on simulated annealing and Pontryagin's minimum principle," *Journal of Power Sources*, vol. 272, pp. 160–168, 2014.
- [22] X. Hu, C. M. Martinez, and Y. Yang, "Charging, power management, and battery degradation mitigation in plug-in hybrid electric vehicles: a unified cost-optimal approach," *Mechanical Systems and Signal Processing*, vol. 87, pp. 4–16, 2017.
- [23] Y. Yang, X. Hu, H. Pei, and Z. Peng, "Comparison of powersplit and parallel hybrid powertrain architectures with a single electric machine: dynamic programming approach," *Applied Energy*, vol. 168, pp. 683–690, 2016.
- [24] S. Xie, X. Hu, S. Qi, and K. Lang, "An artificial neural network-enhanced energy management strategy for plug-in hybrid electric vehicles," *Energy*, vol. 163, pp. 837–848, 2018.
- [25] P. Chambon, S. Curran, S. Huff et al., "Development of a range-extended electric vehicle powertrain for an integrated energy systems research printed utility vehicle," *Applied Energy*, vol. 191, pp. 99–110, 2017.
- [26] R. Xiong, J. Cao, and Q. Yu, "Reinforcement learning-based real-time power management for hybrid energy storage system in the plug-in hybrid electric vehicle," *Applied En*ergy, vol. 211, pp. 538–548, 2018.
- [27] Y. Huang, H. Wang, A. Khajepour et al., "A review of power management strategies and component sizing methods for hybrid vehicles," *Renewable and Sustainable Energy Reviews*, vol. 96, pp. 132–144, 2018.
- [28] J. Peng, H. He, and R. Xiong, "Study on energy management strategies for series-parallel plug-in hybrid electric buses," *Energy Procedia*, vol. 75, pp. 1926–1931, 2015.
- [29] S. J. Moura, J. L. Stein, and H. K. Fathy, "Battery-health conscious power management in plug-in hybrid electric vehicles via electrochemical modeling and stochastic

control," *IEEE Transactions on Control Systems Technology*, vol. 21, no. 3, pp. 679–694, 2013.

- [30] S. Di Cairano, W. Liang, I. V. Kolmanovsky, M. L. Kuang, and A. M. Phillips, "Power smoothing energy management and its application to a series hybrid powertrain," *IEEE Transactions on Control Systems Technology*, vol. 21, no. 6, pp. 2091–2103, 2013.
- [31] M. Ye, X. Song, R. Xiong, and F. Sun, "A novel dynamic performance analysis and evaluation model of series-parallel connected battery pack for electric vehicles," *IEEE Access*, vol. 7, pp. 14256–14265, 2019.
- [32] S. Zhou, P. Walker, and N. Zhang, "Parametric design and regenerative braking control of a parallel hydraulic hybrid vehicle," *Mechanism and Machine Theory*, vol. 146, Article ID 103714, 2020.
- [33] Z. Yuan, L. Teng, S. Fengchun, and H. Peng, "Comparative study of dynamic programming and pontryagin's minimum principle on energy management for a parallel hybrid electric vehicle," *Energies*, vol. 6, no. 4, pp. 2305–2318, 2013.
- [34] D. Liu and H. Li, "A ZVS Bi-directional DC-DC converter for multiple energy storage elements," *IEEE Transactions on Power Electronics*, vol. 21, no. 5, pp. 1513–1517, 2006.
- [35] S. Liu, X. Zhang, H. Guo, and J. Xie, "Multiport DC/DC converter for stand-alone photovoltaic lighting system with battery storage," in *Proceedings of the 2010 International Conference on Electrical and Control Engineering*, pp. 3894–3897, Wuhan, China, June 2010.
- [36] S. Zhang and R. Xiong, "Adaptive energy management of a plug-in hybrid electric vehicle based on driving pattern recognition and dynamic programming," *Applied Energy*, vol. 155, pp. 68–78, 2015.
- [37] A. Solouk, M. Shakiba-herfeh, K. Kannan, H. Solmaz, P. Dice, M. Bidarvatan et al., *Fuel Economy Benefits of Integrating a Multi-Mode Low Temperature Combustion (LTC) Engine in a Series Extended Range Electric Powertrain*, SAE International, Warrendale, PA, USA, 2016.
- [38] L. Xu, J. Li, J. Hua, X. Li, and M. Ouyang, "Optimal vehicle control strategy of a fuel cell/battery hybrid city bus," *International Journal of Hydrogen Energy*, vol. 34, no. 17, pp. 7323–7333, 2009.
- [39] D. Cesiel and C. Zhu, "A closer look at the on-board charger: the development of the second-generation module for the chevrolet volt," *IEEE Electrification Magazine*, vol. 5, no. 1, pp. 36–42, 2017.
- [40] J. Jeong, W. Lee, N. Kim, K. Stutenberg, and A. Rousseau, Control Analysis and Model Validation for BMW 13 Range Extender, SAE International, Warrendale, PA, USA, 2017.
- [41] Ford, Ford New Ford transit and Tourneo custom plug-in hybrids deliver zero emission driving with no range anxiety, SAE International, Warrendale, PA, USA, 2019.
- [42] M. Redelbach, E. D. Özdemir, and H. E. Friedrich, "Optimizing battery sizes of plug-in hybrid and extended range electric vehicles for different user types," *Energy Policy*, vol. 73, pp. 158–168, 2014.
- [43] M. M. Sabri, K. A. Danapalasingam, and M. F. Rahmat, "A review on hybrid electric vehicles architecture and energy management strategies," *Renewable and Sustainable Energy Reviews*, vol. 53, pp. 1433–1442, 2016.
- [44] J. Li, X. Jin, and R. Xiong, "Multi-objective optimization study of energy management strategy and economic analysis for a range-extended electric bus," *Applied Energy*, vol. 194, pp. 798–807, 2017.
- [45] M. Yao, B. Zhu, and N. Zhang, "Adaptive real-time optimal control for energy management strategy of extended range

electric vehicle," *Energy Conversion and Management*, vol. 234, Article ID 113874, 2021.

- [46] J. Li, Y. Wang, J. Chen, and X. Zhang, "Study on energy management strategy and dynamic modeling for auxiliary power units in range-extended electric vehicles," *Applied Energy*, vol. 194, pp. 363–375, 2017.
- [47] M. Rodrigues, S. King, D. Scott, and D. Wang, Advanced Energy Management Strategies for Range Extended Electric Vehicle, SAE International, Warrendale, PA, USA, 2015.
- [48] H. Zhang, L. Fu, J. Song, and Q. Yang, "Power energy management and control strategy study for extended-range auxiliary power unit," *Energy Procedia*, vol. 104, pp. 32–37, 2016.
- [49] Y. L. Guo, X. D. Wang, and J. Long, "The study on 45KW range extended electric vehicle auxiliary power unit," *Mechatronics Engineering Computing and Information Technology*, vol. 556, pp. 2128–2132, 2014.
- [50] B.-C. Chen, Y.-Y. Wu, and H.-C. Tsai, "Design and analysis of power management strategy for range extended electric vehicle using dynamic programming," *Applied Energy*, vol. 113, pp. 1764–1774, 2014.
- [51] L. Zhang, X. Hu, Z. Wang, F. Sun, and D. G. Dorrell, "A review of super-capacitor modeling, estimation, and applications: a control/management perspective," *Renewable and Sustainable Energy Reviews*, vol. 81, pp. 1868–1878, 2018.
- [52] S. Zhang, R. Xiong, and F. Sun, "Model predictive control for power management in a plug-in hybrid electric vehicle with a hybrid energy storage system," *Applied Energy*, vol. 185, pp. 1654–1662, 2017.
- [53] S. Nazari, J. Siegel, and A. Stefanopoulou, "Optimal energy management for a mild hybrid vehicle with electric and hybrid engine boosting systems," *IEEE Transactions on Vehicular Technology*, vol. 68, no. 4, pp. 3386–3399, 2019.
- [54] K. Song, J. Zhang, and T. Zhang, "Design and development of a pluggable PEMFC extended range electric vehicle," in *Proceedings of the 2011 Second International Conference on Mechanic Automation and Control Engineering*, pp. 1144– 1147, Inner Mongolia, China, July 2011.
- [55] T. F. Xian, C. M. Soon, S. Rajoo, and A. Romagnoli, "A parametric study: the impact of components sizing on range extended electric vehicle's driving range," in *Proceedings of the 2016 Asian Conference on Energy, Power and Transportation Electrification (ACEPT)*, pp. 1–6, Singapore, October 2016.
- [56] T.-Q. Tang, L. Chen, S.-C. Yang, and H.-Y. Shang, "An extended car-following model with consideration of the electric vehicle's driving range," *Physica A: Statistical Mechanics and Its Applications*, vol. 430, pp. 148–155, 2015.
- [57] X. Xu, M. Lv, Z. Chen, W. Ji, and R. Gao, "Analysis of electric vehicle extended range misalignment based on rigid-flexible dynamics," *IOP Conference Series: Earth and Environmental Science*, vol. 61, Article ID 012018, 2017.
- [58] P. Spichartz, P. Dost, and C. Sourkounis, "Utilisation of battery electric vehicles and extended range electric vehicles in a field test," in *Proceedings of the 2015 International Conference on Renewable Energy Research and Applications* (*ICRERA*), pp. 1168–1173, Palermo, Italy, November 2015.
- [59] J. C. Guan, B. C. Chen, and Y. Y. Wu, "Design of an adaptive power management strategy for range extended electric vehicles," *Energies*, vol. 12, no. 9, p. 1610, 2019.
- [60] L. Shi, Y. Guo, D. Xiao, Z. Han, and X. Zhou, "Design, optimization, and study of a rare-earth permanent-magnet generator with new consequent-pole rotor for extended-

range electric vehicle," *IEEJ Transactions on Electrical and Electronic Engineering*, vol. 14, no. 6, pp. 917–923, 2019.

- [61] J. Du, J. Chen, M. Gao, and J. Wang, "Energy efficiency oriented design method of power management strategy for range-extended electric vehicles," *Mathematical Problems in Engineering*, vol. 2016, Article ID 9203081, 9 pages, 2016.
- [62] P. Sneha Angeline and M. Newlin Rajkumar, "Evolution of electric vehicle and its future scope," *Materials Today: Proceedings*, vol. 33, pp. 3930–3936, 2020.
- [63] P. Lijewski, M. Kozak, P. Fuć, Ł. Rymaniak, and A. Ziółkowski, "Exhaust emissions generated under actual operating conditions from a hybrid vehicle and an electric one fitted with a range extender," *Transportation Research Part D: Transport and Environment*, vol. 78, Article ID 102183, 2020.
- [64] R. Guo, C. Cao, Y. Mi, and Y. Huang, "Experimental investigation of the noise, vibration and harshness performances of a range-extended electric vehicle," *Proceedings of the Institution of Mechanical Engineers - Part D: Journal of Automobile Engineering*, vol. 230, no. 5, pp. 650–663, 2016.
- [65] L. Ramotar, G. L. Rohrauer, R. Filion, and K. MacDonald, "Experimental verification of a thermal equivalent circuit dynamic model on an extended range electric vehicle battery pack," *Journal of Power Sources*, vol. 343, pp. 383–394, 2017.
- [66] K. Vatanparvar, S. Faezi, I. Burago, M. Levorato, and M. A. Al Faruque, "Extended range electric vehicle with driving behavior estimation in energy management," *IEEE Transactions on Smart Grid*, vol. 10, no. 3, pp. 2959–2968, 2019.
- [67] X. Yan, J. Fleming, and R. Lot, "Extending the range of plugin hybrid electric vehicles by CVT transmission optimal management," *Energy Procedia*, vol. 151, pp. 17–22, 2018.
- [68] T. Y. Kim and S. H. Lee, "Combustion and emission characteristics of wood pyrolysis oil-butanol blended fuels in a DI diesel engine," *International Journal of Automotive Technology*, vol. 16, no. 6, pp. 903–912, 2015.
- [69] G.-S. Lee, D.-H. Kim, J.-H. Han, M.-H. Hwang, and H.-R. Cha, "Optimal operating point determination method design for range-extended electric vehicles based on real driving tests," *Energies*, vol. 12, no. 5, p. 845, 2019.
- [70] H. Liu, Y. Lei, Y. Fu, and X. Li, "Parameter matching and optimization for power system of range-extended electric vehicle based on requirements," *Proceedings of the Institution of Mechanical Engineers Part D Journal of Automobile Engineering*, vol. 234, no. 14, pp. 3316–3328, 2020.
- [71] Y. Jiang, C. Lin, and W. Cao, "Parameters optimization for extended-range electric vehicle based on improved chaotic particle swarm optimization," *International Journal of Grid* and Distributed Computing, vol. 9, pp. 1–10, 2016.
- [72] A. Kopczyński, P. Krawczyk, and J. Lasocki, "Parameters selection of extended-range electric vehicle supplied with alternative fuel," *E3S Web Conference*, vol. 44, p. 00073, 2018.
- [73] G. Ren, S. Shan, G. Ma, X. Shang, S. Zhu, and Q. Zhang, "Review of energy storage technologies for extended range electric vehicle," *Journal of Applied Science and Engineering Section A*, vol. 22, pp. 69–82, 2019.
- [74] D. Baek and N. Chang, "Runtime power management of battery electric vehicles for extended range with consideration of driving time," *IEEE Transactions on Very Large Scale Integration Systems*, vol. 27, no. 3, pp. 549–559, 2019.
- [75] J. Zhao, B. Han, and S. Bei, "Start-stop moment optimization of range extender and control strategy design for extended-

range electric vehicle," *IOP Conference Series: Materials Science and Engineering*, vol. 241, Article ID 012025, 2017.

- [76] D. Wu and L. Feng, "On-off control of range extender in extended-range electric vehicle using bird swarm intelligence," *Electronics*, vol. 8, no. 11, p. 1223, 2019.
- [77] J. Hall, H. Marlok, M. Bassett, and M. Warth, "Analysis of real world data from a range extended electric vehicle demonstrator," SAE International Journal of Alternative Powertrains, vol. 4, no. 1, pp. 20–33, 2014.
- [78] J. C. Figueroa-Gar, E. R. López-Santana, J. L. Villa-Ram\'\irez, and R. Ferro-Escobar, "Applied computer sciences in engineering: 4th workshop on engineering applications WEA 2017 cartagena Colombia," *Proceedings*, vol. 742, pp. 27–29, 2017.
- [79] B. Wahono, A. Nur, W. B. Santoso, and A. Praptijanto, "A comparison study of range-extended engines for electric vehicle based on vehicle simulator," *Journal of Mechanical Engineering Science*, vol. 10, no. 1, pp. 1803–1816, 2016.
- [80] X. Wang, H. Lv, Q. Sun, Y. Mi, and P. Gao, "A proportional resonant control strategy for efficiency improvement in extended range electric vehicles," *Energies*, vol. 10, no. 2, p. 204, 2017.
- [81] W. Song, X. Zhang, Y. Tian, and L. Xi, "A charging management-based intelligent control strategy for extendedrange electric vehicles," *Journal of Zhejiang University Science*, vol. 17, no. 11, pp. 903–910, 2016.
- [82] W. Lee, H. Jeoung, D. Park, and N. Kim, "An adaptive concept of PMP-based control for saving operating costs of extended-range electric vehicles," *IEEE Transactions on Vehicular Technology*, vol. 68, pp. 11505–11512, 2019.
- [83] G. Pozzato, S. Formentin, G. Panzani, and S. M. Savaresi, "Least costly energy management for extended-range electric vehicles with noise emissions characterization," *IFAC-PapersOnLine*, vol. 52, no. 5, pp. 586–591, 2019.
- [84] Z. Zhang, X. Zhang, W. Chen et al., "A high-efficiency energy regenerative shock absorber using super-capacitors for renewable energy applications in range extended electric vehicle," *Applied Energy*, vol. 178, pp. 177–188, 2016.
- [85] N. V. Satpute, S. Singh, and S. M. Sawant, "Energy harvesting shock absorber with electromagnetic and fluid damping," *Advances in Mechanical Engineering*, Hindawi Publishing Corporation, London, UK, 2014.
- [86] Y. Zhang, X. Zhang, M. Zhan, K. Guo, F. Zhao, and Z. Liu, "Study on a novel hydraulic pumping regenerative suspension for vehicles," *Journal of the Franklin Institute*, vol. 352, no. 2, pp. 485–499, 2015.
- [87] C. Qiu, G. Wang, M. Meng, and Y. Shen, "A novel control strategy of regenerative braking system for electric vehicles under safety critical driving situations," *Energy*, vol. 149, pp. 329–340, 2018.
- [88] H. Liu, Y. Lei, Y. Fu, and X. Li, "An optimal slip ratio-based revised regenerative braking control strategy of rangeextended electric vehicle," *Energies*, vol. 13, no. 6, p. 1526, 2020.
- [89] X. Zhao, S. Xu, Y. Ye, M. Yu, and G. Wang, "Composite braking AMT shift strategy for extended-range heavy commercial electric vehicle based on LHMM/ANFIS braking intention identification," *Cluster Computing*, vol. 22, no. S4, pp. 8513–8528, 2019.
- [90] M. S. Aygen and M. Inci, "Zero-sequence current injection based power flow control strategy for grid inverter interfaced renewable energy systems," *Energy Sources Part A Recover Util Environ Eff*, vol. 23, pp. 1–22, 2020.

- [91] A. Ozden, S. Shahgaldi, X. Li, and F. Hamdullahpur, "A review of gas diffusion layers for proton exchange membrane fuel cells—with a focus on characteristics, characterization techniques, materials and designs," *Progress in Energy and Combustion Science*, vol. 74, pp. 50–102, 2019.
- [92] M. İnci, "Active reactive energy control scheme for gridconnected fuel cell system with local inductive loads," *Energy*, vol. 197, Article ID 117191, 2020.
- [93] Y. Zhou, H. Li, A. Ravey, and M.-C. Péra, "An integrated predictive energy management for light-duty rangeextended plug-in fuel cell electric vehicle," *Journal of Power Sources*, vol. 451, Article ID 227780, 2020.
- [94] M. İnci, M. Büyük, M. H. Demir, and G. İlbey, "A review and research on fuel cell electric vehicles: topologies, power electronic converters, energy management methods, technical challenges, marketing and future aspects," *Renewable and Sustainable Energy Reviews*, vol. 137, Article ID 110648, 2021.
- [95] Y. Chu, Y. Wu, J. Chen, S. Zheng, and Z. Wang, "Design of energy and materials for ammonia-based extended-range electric vehicles," *Energy Procedia*, vol. 158, pp. 3064–3069, 2019.
- [96] Z. Dimitrova and F. Maréchal, "Environomic design for electric vehicles with an integrated solid oxide fuel cell (SOFC) unit as a range extender," *Renewable Energy*, vol. 112, pp. 124–142, 2017.
- [97] M. U. Karaoğlan, N. S. Kuralay, and C. O. Colpan, "Investigation of the effects of battery types and power management algorithms on drive cycle simulation for a rangeextended electric vehicle powertrain," *International Journal* of Green Energy, vol. 16, pp. 1–11, 2019.
- [98] Y. Liu, J. Li, Z. Chen, D. Qin, and Y. Zhang, "Research on a multi-objective hierarchical prediction energy management strategy for range extended fuel cell vehicles," *Journal* of *Power Sources*, vol. 429, pp. 55–66, 2019.
- [99] C. Geng, X. Jin, and X. Zhang, "Simulation research on a novel control strategy for fuel cell extended-range vehicles," *International Journal of Hydrogen Energy*, vol. 44, no. 1, pp. 408–420, 2019.
- [100] A. Almasi, "Bright future of micro-turbines: nuclear emergency cooling and miniature generator," *Australian Journal of Mechanical Engineering*, vol. 11, no. 2, pp. 169– 175, 2013.
- [101] A. Karvountzis-Kontakiotis, A. M. Andwari, A. Pesyridis, S. Russo, R. Tuccillo, and V. Esfahanian, "Application of micro gas turbine in range-extended electric vehicles," *Energy*, vol. 147, pp. 351–361, 2018.
- [102] W. S. Bou Nader, C. J. Mansour, and M. G. Nemer, "Optimization of a Brayton external combustion gas-turbine system for extended range electric vehicles," *Energy*, vol. 150, pp. 745–758, 2018.
- [103] A. Reine and W. Bou Nader, "Fuel consumption potential of different external combustion gas-turbine thermodynamic configurations for extended range electric vehicles," *Energy*, vol. 175, pp. 900–913, 2019.
- [104] M. S. ElNozahy and M. M. A. Salama, "Studying the feasibility of charging plug-in hybrid electric vehicles using photovoltaic electricity in residential distribution systems," *Electric Power Systems Research*, vol. 110, pp. 133–143, 2014.
- [105] W. Khan, A. Ahmad, F. Ahmad, and M. Saad Alam, "A comprehensive review of fast charging infrastructure for electric vehicles," *Smart Science*, vol. 6, pp. 1–15, 2018.
- [106] S. Khan, A. Ahmad, F. Ahmad, M. Shafaati Shemami, M. Saad Alam, and S. Khateeb, "A comprehensive review on

solar powered electric vehicle charging system," *Smart Science*, vol. 6, no. 1, pp. 54–79, 2018.

- [107] F. Ahmad, M. Khalid, and B. K. Panigrahi, "Development in energy storage system for electric transportation: a comprehensive review," *Journal of Energy Storage*, vol. 43, Article ID 103153, 2021.
- [108] F. Ahmad, M. S. Alam, and M. Asaad, "Developments in xEVs charging infrastructure and energy management system for smart microgrids including xEVs," *Sustainable Cities and Society*, vol. 35, pp. 552–564, 2017.
- [109] M. Khalid, F. Ahmad, B. K. Panigrahi, and L. Al-Fagih, "A comprehensive review on advanced charging topologies and methodologies for electric vehicle battery," *Journal of Energy Storage*, vol. 53, Article ID 105084, 2022.
- [110] R. Keshri, M. Bertoluzzo, and G. Buja, "Integration of a photovoltaic panel with an electric city car," *Electric Power Components and Systems*, vol. 42, no. 5, pp. 481–495, 2014.
- [111] M. F. Ezzat and I. Dincer, "Development, analysis and assessment of a fuel cell and solar photovoltaic system powered vehicle," *Energy Conversion and Management*, vol. 129, pp. 284–292, 2016.
- [112] J. Li and J. Zhao, "Energy recovery for hybrid hydraulic excavators: flywheel-based solutions," *Automation in Construction*, vol. 125, Article ID 103648, 2021.
- [113] F. Khoshnoud, Y. Zhang, R. Shimura et al., "Energy regeneration from suspension dynamic modes and selfpowered actuation," *IEEE*, vol. 20, no. 5, pp. 2513–2524, 2015.
- [114] A. Maravandi and M. Moallem, "Regenerative shock absorber using a two-leg motion conversion mechanism," *IEEE*, vol. 20, no. 6, pp. 2853–2861, 2015.
- [115] Z. Li, L. Zuo, G. Luhrs, L. Lin, and Y. Qin, "Electromagnetic energy-harvesting shock absorbers: design, modeling, and road tests," *IEEE Transactions on Vehicular Technology*, vol. 62, no. 3, pp. 1065–1074, 2013.
- [116] W. B. Nader, J. Chamoun, and C. Dumand, "Optimization of the thermodynamic configurations of a thermoacoustic engine auxiliary power unit for range extended hybrid electric vehicles," *Energy*, vol. 195, Article ID 116952, 2020.
- [117] B. Chen, A. A. Yousif, P. H. Riley, and D. B. Hann, "Development and Assessment of Thermoacoustic Generators Operating by Waste Heat from Cooking Stove," *Engineering*, vol. 4, p. 25859, 2012, https://www.scirp.org/journal/ PaperInformation.aspx?PaperID=25859.
- [118] S. Backhaus and G. W. Swift, "A thermoacoustic-Stirling heat engine: detailed study," *Journal of the Acoustical Society* of America, vol. 107, no. 6, pp. 3148–3166, 2000.
- [119] S. Hahn, J. Braun, H. Kemmer, and H.-C. Reuss, "Optimization of the efficiency and degradation rate of an automotive fuel cell system," *International Journal of Hydrogen Energy*, vol. 46, pp. 29459–29477, 2021.
- [120] J. D. Boyes and N. H. Clark, "Technologies for energy storage. Flywheels and super conducting magnetic energy storage," in *Proceedings of the 2000 Power Engineering Society Summer Meeting (Cat. No.00CH37134)*, vol. 3, pp. 1548–1550, Seattle, WA, USA, July 2000.
- [121] W. Wang, Y. Li, M. Shi, and Y. Song, "Optimization and control of battery-flywheel compound energy storage system during an electric vehicle braking," *Energy*, vol. 226, Article ID 120404, 2021.
- [122] S. Vazquez, S. M. Lukic, E. Galvan, L. G. Franquelo, and J. M. Carrasco, "Energy storage systems for transport and grid applications," *IEEE Transactions on Industrial Electronics*, vol. 57, no. 12, pp. 3881–3895, 2010.

- [123] D. Erdemir and I. Dincer, "Assessment of renewable energydriven and flywheel integrated fast-charging station for electric buses: a case study," *Journal of Energy Storage*, vol. 30, Article ID 101576, 2020.
- [124] P. Puddu and M. Paderi, "Hydro-pneumatic accumulators for vehicles kinetic energy storage: influence of gas compressibility and thermal losses on storage capability," *Energy*, vol. 57, pp. 326–335, 2013.
- [125] M. Eolo, A Wind-Powered EV from Colombia N.d, ASME, Washington, DC, USA, 2022.
- [126] H. Fathabadi, "Recovering waste vibration energy of an automobile using shock absorbers included magnet movingcoil mechanism and adding to overall efficiency using wind turbine," *Energy*, vol. 189, Article ID 116274, 2019.
- [127] D. S. Puma-Benavides, J. Izquierdo-Reyes, J. D. Calderon-Najera, and R. A. Ramirez-Mendoza, "A systematic review of technologies, control methods, and optimization for extended-range electric vehicles," *Applied Sciences*, vol. 11, no. 15, p. 7095, 2021.
- [128] H. Fathabadi, "Utilizing solar and wind energy in plug-in hybrid electric vehicles," *Energy Conversion and Management*, vol. 156, pp. 317–328, 2018.
- [129] F. Ji, X. Zhang, F. Du et al., "Experimental and numerical investigation on micro gas turbine as a range extender for electric vehicle," *Applied Thermal Engineering*, vol. 173, Article ID 115236, 2020.
- [130] J. Duan, S. Fan, Q. An, L. Sun, and G. Wang, "A comparison of micro gas turbine operation modes for optimal efficiency based on a nonlinear model," *Energy*, vol. 134, pp. 400–411, 2017.
- [131] A. A. Barakat, J. H. Diab, N. S. Badawi, W. S. Bou Nader, and C. J. Mansour, "Combined cycle gas turbine system optimization for extended range electric vehicles," *Energy Conversion and Management*, vol. 226, Article ID 113538, 2020.
- [132] F. F. Nicolosi and M. Renzi, "Effect of the regenerator efficiency on the performance of a micro gas turbine fed with alternative fuels," *Energy Procedia*, vol. 148, pp. 687–694, 2018.
- [133] M. F. Ezzat and I. Dincer, "Development and exergetic assessment of a new hybrid vehicle incorporating gas turbine as powering option," *Energy*, vol. 170, pp. 112–119, 2019.
- [134] Z. Junzhi, L. Yutong, L. Chen, and Y. Ye, "New regenerative braking control strategy for rear-driven electrified minivans," *Energy Conversion and Management*, vol. 82, pp. 135–145, 2014.
- [135] J. Zhao, Y. Ma, Z. Zhang, S. Wang, and S. Wang, "Optimization and matching for range-extenders of electric vehicles with artificial neural network and genetic algorithm," *Energy Conversion and Management*, vol. 184, pp. 709–725, 2019.
- [136] Y. Yang, Q. He, Y. Chen, and C. Fu, "Efficiency optimization and control strategy of regenerative braking system with dual motor," *Energies*, vol. 13, no. 3, p. 711, 2020.
- [137] G. A. Sadiq, R. Al-Dadah, and S. Mahmoud, "Development of rotary Wankel devices for hybrid automotive applications," *Energy Conversion and Management*, vol. 202, Article ID 112159, 2019.
- [138] C. Ji, H. Wang, C. Shi, S. Wang, and J. Yang, "Multi-objective optimization of operating parameters for a gasoline Wankel rotary engine by hydrogen enrichment," *Energy Conversion and Management*, vol. 229, Article ID 113732, 2021.

- [139] J. J. G. Martins, K. Uzuneanu, B. S. Ribeiro, and O. Jasansky, "Thermodynamic analysis of an over-expanded engine," SAE Transactions, vol. 56, pp. 476–490, 2004.
- [140] M. R. Range Extender, Mazda 2 RE range-extender: return of the rotary-news-car and driver, Hearst Autos Inc, Hearst, Canada.
- [141] T. Ouyang, Z. Zhao, Z. Wang, M. Zhang, and B. Liu, "A highefficiency scheme for waste heat harvesting of solid oxide fuel cell integrated homogeneous charge compression ignition engine," *Energy*, vol. 229, Article ID 120720, 2021.
- [142] T. Toyota, "Second Generation Toyota Mirai. Still Zero Emissions," 2019, https://www.toyota-europe.com/worldoftoyota/%20articles-news-events/2019/new-mirai-concept %202019.
- [143] T. Taner, "The novel and innovative design with using H2 fuel of PEM fuel cell: efficiency of thermodynamic analyze," *Fuel*, vol. 302, Article ID 121109, 2021.
- [144] H. Hyundai, "NEXO de pila de combustible|Hyundai| Hyundai Motor España," 2023, https://www.hyundai.com/ es/%20modelos/nexo.html%20n.d.
- [145] J. Hoeflinger and P. Hofmann, "Air mass flow and pressure optimisation of a PEM fuel cell range extender system," *International Journal of Hydrogen Energy*, vol. 45, no. 53, pp. 29246–29258, 2020.
- [146] Nikola Motor Co, "Nikola Two 2021," 2021, https:// nikolamotor.com/two.
- [147] F. Leach, G. Kalghatgi, R. Stone, and P. Miles, "The scope for improving the efficiency and environmental impact of internal combustion engines," *Transport Engineer*, vol. 1, Article ID 100005, 2020.
- [148] G. Menzel, S. Hennings Och, V. Cocco Mariani, L. Mauro Moura, and E. Domingues, "Multi-objective optimization of the volumetric and thermal efficiencies applied to a multicylinder internal combustion engine," *Energy Conversion* and Management, vol. 216, Article ID 112930, 2020.
- [149] BMW, "BMW I3s 2020," 2020, https://www.guideautoweb. com/en/makes/bmw/i3/2020/specifications/%0As-rex/.
- [150] G. Motors, "Chevrolet Volt Brochure nd," 2019, https:// www.chevrolet.ca/content/dam/chevrolet/na/%0Acanada/ english/index/download-a-brochure/02-pdfs/my19-cdn-voltbrochure-en.pdf.
- [151] The New BMW I3 the New B I3s, "The_new_BMW_i3_LCI_ BMW_i3s_Specifications," 2017, https://www. bimmerarchive.org/article/2017-08-29-the-new-bmw-i3-thenew-bmw-i3s.html.
- [152] A. Solouk and M. Shahbakhti, "Modelling and energy management of an HCCI-based powertrain for series hybrid and extended range electric vehicles," *International Journal* of Powertrains, vol. 6, no. 3, pp. 226–258, 2017.
- [153] H. H. Eldeeb, A. T. Elsayed, C. R. Lashway, and O. Mohammed, "Hybrid energy storage sizing and power splitting optimization for plug-in electric vehicles," *IEEE Transactions on Industry Applications*, vol. 55, no. 3, pp. 2252–2262, 2019.
- [154] J. Bauman and M. Kazerani, "An analytical optimization method for improved fuel cell-battery-ultracapacitor powertrain," *IEEE Transactions on Vehicular Technology*, vol. 58, no. 7, pp. 3186–3197, 2009.
- [155] P. Bentley, D. A. Stone, and N. Schofield, "The parallel combination of a VRLA cell and super-capacitor for use as a hybrid vehicle peak power buffer," *Journal of Power Sources*, vol. 147, no. 1-2, pp. 288–294, 2005.
- [156] J. Moreno, M. E. Ortuzar, and J. W. Dixon, "Energymanagement system for a hybrid electric vehicle, using

ultracapacitors and neural networks," *IEEE Transactions on Industrial Electronics*, vol. 53, no. 2, pp. 614–623, 2006.

- [157] H. Friedl, G. Fraidl, C. Hubmann, H. Sorger, G. Teuschl, and C. Martin, "Range Extender Technology for Electric Vehicles," in *Proceedings of the 2018 5th International Conference* on Electric Vehicular Technology (ICEVT), pp. 1–8, Surakarta, Indonesia, October 2018.
- [158] J. Andert, K. Herold, R. Savelsberg, and M. Pischinger, "NVH optimization of range extender engines by electric torque profile shaping," *IEEE Transactions on Control Systems Technology*, vol. 25, no. 4, pp. 1465–1472, 2017.
- [159] M. Borghi, E. Mattarelli, J. Muscoloni, C. A. Rinaldini, T. Savioli, and B. Zardin, "Design and experimental development of a compact and efficient range extender engine," *Applied Energy*, vol. 202, pp. 507–526, 2017.
- [160] R. D. Reitz and G. Duraisamy, "Review of high efficiency and clean reactivity controlled compression ignition (RCCI) combustion in internal combustion engines," *Progress in Energy and Combustion Science*, vol. 46, pp. 12–71, 2015.
- [161] R. Guo, M. Wang, and C. Cao, "Gas torque compensation control for range extender start-stop vibration reduction in electric vehicle," *Proceedings of the Institution of Mechanical Engineers - Part I: Journal of Systems & Control Engineering*, vol. 231, no. 7, pp. 567–581, 2017.
- [162] Y. Qin, X. Tang, T. Jia et al., "Noise and vibration suppression in hybrid electric vehicles: state of the art and challenges," *Renewable and Sustainable Energy Reviews*, vol. 124, Article ID 109782, 2020.
- [163] Mazda, Mazda Rotary Engine to Return as EV Range-Extender, Hearst Autos Inc, Hearst, Canada, 2018.
- [164] F. V. Conte, "Battery and battery management for hybrid electric vehicles: a review," E & I: Elektrotechnik und Informationstechnik, vol. 123, no. 10, pp. 424-431, 2006.
- [165] L. Lu, X. Han, J. Li, J. Hua, and M. Ouyang, "A review on the key issues for lithium-ion battery management in electric vehicles," *Journal of Power Sources*, vol. 226, pp. 272–288, 2013.
- [166] W. Zuo, R. Li, C. Zhou, Y. Li, J. Xia, and J. Liu, "Batterysupercapacitor hybrid devices: recent progress and future prospects," *Advanced Science*, vol. 4, no. 7, pp. 1600539– 1600621, 2017.
- [167] K. T. Chau, Y. S. Wong, and C. C. Chan, "An overview of energy sources for electric vehicles," *Energy Conversion and Management*, vol. 40, no. 10, pp. 1021–1039, 1999.
- [168] L. Sun and N. Zhang, "Design, implementation and characterization of a novel bi-directional energy conversion system on DC motor drive using super-capacitors," *Applied Energy*, vol. 153, pp. 101–111, 2015.
- [169] L. Kouchachvili, W. Yaïci, and E. Entchev, "Hybrid battery/ super-capacitor energy storage system for the electric vehicles," *Journal of Power Sources*, vol. 374, pp. 237–248, 2018.
- [170] B. Xiao, H. Lu, H. Wang, J. Ruan, and N. Zhang, "Enhanced regenerative braking strategies for electric vehicles: dynamic performance and potential analysis," *Energies*, vol. 10, no. 11, p. 1875, 2017.
- [171] Z. Song, J. Li, X. Han et al., "Multi-objective optimization of a semi-active battery/super-capacitor energy storage system for electric vehicles," *Applied Energy*, vol. 135, pp. 212–224, 2014.
- [172] T. Li, L. Huang, and H. Liu, "Energy management and economic analysis for a fuel cell super-capacitor excavator," *Energy*, vol. 172, pp. 840–851, 2019.
- [173] X. Hu, T. Liu, X. Qi, and M. Barth, "Reinforcement learning for hybrid and plug-in hybrid electric vehicle energy

management: recent advances and prospects," IEEE Industrial Electronics Magazine, vol. 13, no. 3, pp. 16–25, 2019.

- [174] M. S. Khan, I. Ahmad, H. Armaghan, and N. Ali, "Backstepping sliding mode control of FC-UC based hybrid electric vehicle," *IEEE Access*, vol. 6, pp. 77202–77211, 2018.
- [175] N. O'Connell, A. Röll, R. Lechner, T. Luo, and M. Brautsch, "PODE-blend as pilot fuel in a biomethane dual fuel engine: experimental analysis of performance, combustion and emissions characteristics," *Renewable Energy*, vol. 143, pp. 101–111, 2019.
- [176] A. K. Agarwal, A. P. Singh, and R. K. Maurya, "Evolution, challenges and path forward for low temperature combustion engines," *Progress in Energy and Combustion Science*, vol. 61, pp. 1–56, 2017.
- [177] Z. Salameh, *Renewable Energy System Design*, Academic Press, Cambridge, MS, USA, 2014.
- [178] H. L. P. Contact, K. Stutenberg, and S. C. Ave, *Toyota Mirai Testing*, United States Department of Energy, Washington, DC, USA, 2018.
- [179] X. Han, F. Li, T. Zhang, T. Zhang, and K. Song, "Economic energy management strategy design and simulation for a dual-stack fuel cell electric vehicle," *International Journal* of Hydrogen Energy, vol. 42, no. 16, pp. 11584–11595, 2017.
- [180] S. Ahmadi, S. M. T. Bathaee, and A. H. Hosseinpour, "Improving fuel economy and performance of a fuel-cell hybrid electric vehicle (fuel-cell, battery, and ultra-capacitor) using optimized energy management strategy," *Energy Conversion and Management*, vol. 160, pp. 74–84, 2018.
- [181] N. Bizon, "Real-time optimization strategy for fuel cell hybrid power sources with load-following control of the fuel or air flow," *Energy Conversion and Management*, vol. 157, pp. 13–27, 2018.
- [182] L. Barelli, G. Bidini, and A. Ottaviano, "Optimization of a PEMFC/battery pack power system for a bus application," *Applied Energy*, vol. 97, pp. 777–784, 2012.
- [183] X. Zhang, Z. Wu, X. Hu, W. Qian, and Z. Li, "Trajectory optimization-based auxiliary power unit control strategy for an extended range electric vehicle," *IEEE Transactions on Vehicular Technology*, vol. 66, no. 12, pp. 10866–10874, 2017.
- [184] X. Ba, Y. Guo, J. Zhu, and C. Zhang, "An equivalent circuit model for predicting the core loss in a claw-Pole permanent magnet motor with soft magnetic composite core," *IEEE Transactions on Magnetics*, vol. 54, no. 11, pp. 1–6, 2018.
- [185] W. Gao, "Performance comparison of a fuel cell-battery hybrid powertrain and a fuel cell-ultracapacitor hybrid powertrain," *IEEE Transactions on Vehicular Technology*, vol. 54, no. 3, pp. 846–855, 2005.
- [186] M. Zandi, A. Payman, J.-P. Martin, S. Pierfederici, B. Davat, and F. Meibody-Tabar, "Energy management of a fuel cell/ super-capacitor/battery power source for electric vehicular applications," *IEEE Transactions on Vehicular Technology*, vol. 60, no. 2, pp. 433–443, 2011.
- [187] M. G. Simoes, B. Blunier, and A. Miraoui, "Fuzzy-based energy management control: design of a battery auxiliary power unit for remote applications," *IEEE Industry Applications Magazine*, vol. 20, no. 4, pp. 41–49, 2014.
- [188] X. Zhao and S. Niu, "Design and optimization of a novel slot-PM-assisted variable flux reluctance generator for hybrid electric vehicles," *IEEE Transactions on Energy Conversion*, vol. 33, no. 4, pp. 2102–2111, 2018.
- [189] M. Rogge, S. Rothgang, and D. U. Sauer, "Operating Strategies for a Range Extender Used in Battery Electric Vehicles," in *Proceedings of the 2013 IEEE Vehicle Power and*

Propulsion Conference (VPPC), pp. 1–5, Beijing, China, October 2013.

- [190] Y. Yan, Q. Li, W. Chen, B. Su, J. Liu, and L. Ma, "Optimal energy management and control in multimode equivalent energy consumption of fuel cell/supercapacitor of hybrid electric tram," *IEEE Transactions on Industrial Electronics*, vol. 66, no. 8, pp. 6065–6076, 2019.
- [191] R. Verma and P. A. Lakshminiarayanan, "A case study on the application of a genetic algorithm for optimization of engine parameters," *Proceedings of the Institution of Mechanical Engineers - Part D: Journal of Automobile Engineering*, vol. 220, no. 4, pp. 471–479, 2006.
- [192] L. Xi, X. Zhang, C. Sun, Z. Wang, X. Hou, and J. Zhang, "Intelligent energy management control for extended range electric vehicles based on dynamic programming and neural network," *Energies*, vol. 10, no. 11, p. 1871, 2017.
- [193] S. Di Cairano, D. Bernardini, A. Bemporad, and I. V. Kolmanovsky, "Stochastic MPC with learning for driver-predictive vehicle control and its application to HEV energy management," *IEEE Transactions on Control Systems Technology*, vol. 22, no. 3, pp. 1018–1031, 2014.
- [194] T. Liu, X. Hu, W. Hu, and Y. Zou, "A heuristic planning reinforcement learning-based energy management for power-split plug-in hybrid electric vehicles," *IEEE Transactions on Industrial Informatics*, vol. 15, no. 12, pp. 6436– 6445, 2019.
- [195] R. Lian, H. Tan, J. Peng, Q. Li, and Y. Wu, "Cross-type transfer for deep reinforcement learning based hybrid electric vehicle energy management," *IEEE Transactions on Vehicular Technology*, vol. 69, no. 8, pp. 8367–8380, 2020.
- [196] B. Xu, X. Hu, X. Tang et al., "Ensemble reinforcement learning-based supervisory control of hybrid electric vehicle for fuel economy improvement," *IEEE Transactions on Transportation Electrification*, vol. 6, no. 2, pp. 717–727, 2020.
- [197] C. V. C. N. Chevy, Volt Concept NAIAS 2007, Hearst Autos Inc, Hearst, Canada, 2007.
- [198] U. S. D. Energy, *Chevrolet Volt fuel economy*, Hearst Autos Inc, Hearst, Canada, 2013.
- [199] J. Smart and S. Schey, "Battery electric vehicle driving and charging behavior observed early in the EV project," SAE International Journal of Alternative Powertrains, vol. 1, pp. 27–33, 2012.
- [200] M. Anwar, S. M. N. Hasan, M. Teimor, M. Korich, and M. B. Hayes, "Development of a Power Dense and Environmentally Robust Traction Power Inverter for the Second-Generatio Chevrolet VOLT Extended-Range EV," in Proceedings of the 2015 IEEE Energy Conversion Congress and Exposition (ECCE), pp. 6006–6013, Montreal, Canada, September 2015.
- [201] J. Gonder, E. Wood, and S. Rajagopalan, "Connectivityenhanced route selection and adaptive control for the Chevrolet Volt," *Journal of Traffic and Transportation Engineering*, vol. 4, no. 1, 2016.
- [202] J. Smart, Advanced Vehicle Testing Activity-Ccold Weather On-Road Testing of a 2012 Chevrolet Volt, Idaho National Laboratory, Idaho Falls, ID, USA, 2014.
- [203] A. Loiselle-Lapointe, A. J. Conde, and H. Ribberink, "Chevrolet Volt on-road test programs in Canada part 1: effects of drive cycle, ambient temperature and accessory usage on energy consumption and all-electric range," *International Journal of Automotive Technology*, vol. 18, no. 1, pp. 103–115, 2017.

- [204] B. Xiao, J. Ruan, W. Yang, P. D. Walker, and N. Zhang, "A review of pivotal energy management strategies for extended range electric vehicles," *Renewable and Sustainable Energy Reviews*, vol. 149, Article ID 111194, 2021.
- [205] A. F. Burke, "Batteries and ultracapacitors for electric, hybrid, and fuel cell vehicles," *Proceedings of the IEEE*, vol. 95, no. 4, pp. 806–820, 2007.
- [206] K. Maxwell, Ultra Capacitors 2.7V SERIES Product Specifications, maxwell Technologies, San Diego, CA, USA, 2018.
- [207] Y. Kim, V. Raghunathan, and A. Raghunathan, "Design and management of battery-supercapacitor hybrid electrical energy storage systems for regulation services," *IEEE Transactions on Multi-Scale Computing Systems*, vol. 3, no. 1, pp. 12–24, 2017.
- [208] C. Iclodean, B. Varga, N. Burnete, D. Cimerdean, and B. Jurchiş, "Comparison of different battery types for electric vehicles," *IOP Conference Series: Materials Science and Engineering*, vol. 252, Article ID 012058, 2017.
- [209] S. Mekhilef, R. Saidur, and A. Safari, "Comparative study of different fuel cell technologies," *Renewable and Sustainable Energy Reviews*, vol. 16, no. 1, pp. 981–989, 2012.
- [210] Y. Haseli, "Maximum conversion efficiency of hydrogen fuel cells," *International Journal of Hydrogen Energy*, vol. 43, no. 18, pp. 9015–9021, 2018.
- [211] Q.-M. Luo, "High step-up converter with capacity of multiple input," *IET Power Electronics*, vol. 5, no. 7, pp. 524–531, 2012.
- [212] Bloomberg, Electric Vehicle Outlook, Bloomberg, New York, NY, USA, 2019.
- [213] P. Sharma and T. S. Bhatti, "A review on electrochemical double-layer capacitors," *Energy Conversion and Management*, vol. 51, pp. 2901–2912, 2010.